Author's response to reviews

Title: Effectiveness of second generation antipsychotics: A systematic review of randomized trials.

Authors:

   Erik Johnsen (erij@ihelse.net)
   Hugo A Jørgensen (hugo.jorgensen@psyk.uib.no)

Version: 2 Date: 21 April 2008

Author's response to reviews: see over
Dear Sirs,

We thank you for accepting our manuscript “Effectiveness of second generation antipsychotics: A systematic review of randomized trials” for publication, and hereby submit the revised and final version. The following revisions have, as requested, been made in the final version:

1. The manuscript ID has been removed from the title page.
2. The keywords have been removed.
3. The methods section has been renamed “Methods”.
4. The statement addressing funding has been moved to the Acknowledgements section.
5. The title, figure number and legend have been removed from the figure file, and are integrated in the manuscript after the reference list.
6. The tables are removed from the submission system and are integrated in the manuscript after the reference list. The tables are formatted from cells. The table titles of the individual tables are placed above the table, and the legends below.
7. The additional files 6-9 are removed from the submission system.

Yours sincerely,

Erik Johnsen, M.D.                              Hugo A. Jørgensen. M.D., Ph.D.

Gateadresse: Sandviksleiet 1, N-5035 Bergen, Norge
Postadresse: Postboks 23, Sandviken 5812 Bergen, Norge
Telefon: +4755 95 8556/8438
Telefax: + 4755958436
Email: Hugo.jorgensen@psyk.uib.no